Chapter 12

Blending...

FLOOPS

Blending languages

SWIG - Simplified Wrapper and Interface Generator S/W tool that connects C and C++ with scripting languages Generates wrapper code that scripting languages use to access C/C++.

Perl, Python, Tcl/Tk, Ruby, Java, OCAML, C# ...

Today though...

Perl and Tk C and Tk VRML and Java Java and Tk

Perl and Tk

Have to install the Tk module for perl Download Tk-804.026.tar.gz, and then perl Makefile.pl XFT=1 make make install

GUI

Perl/Tk code

Perl has a Tk module:

Menu bar

Menu items

Edit menu

View menu

Dialog box

Code

http://www.comp.nus.edu.sg/~cs3283/ftp/original.pl

It may be run by typing “perl original.pl”.

C and Tk

There are various ways...

Call C within Tk Call Tk within C Communication

C and Tk

This one is a C program It loads and runs a Tk program along with its interpreter. Tk in turn can call-back C procedures if needed

Viewer

C and Tk

Compilation

On a Win32 system, we compile this as:

gcc -o myexe CplusTclTk.c -mwindows -ltcl80 -ltk80

On a UNIX system we use:

gcc -o myexe CplusTclTk.c -ltk -ltcl -lX11 -lm -ldl

VRML and Java

VRML is a scene description language

“The EAI allows you to control the contents of a VRML browser window embedded in a web page from a Java (tm) applet on the same page.”

http://www.frontiernet.net/~imaging/eaifaq.html

(People poser and Tiny3D are OK but there are problems...)

VRML and Java3DVNT

VRML software

The root.wrl file

VRML nodes

Java

The java software maintains a link to a remote data collector, Uses the EAI to modify the images in the VRML view.

Java 1Java 1

Java/Tk code

As for C.... we have

Java calls to Tk Tk calls to Java Partial implementation of Tk inside Java/Swing

Tk code

Java/Tk code

Tk code

The JavaKit code

Summary of topics

Tk blended with C Tk blended with perl VRML, Java and the EAI Tk blended with Java